Author's response to reviews

Title: Working relationship between nurses and doctors in university teaching hospitals in Southern Nigeria

Authors:

Roseline I Igbimi (ogbirose@yahoo.com)
Clement A Adebamowo (cadebamo@yahoo.com)

Version: 4 Date: 20 December 2005

Author's response to reviews:

The focus of the article is identification of factors that affect the working relationship between doctors and nurses in Nigeria. The authors wish to identify which of these factors are significant and suggest ways of improving this relationship so that health care delivery can be more efficient.

The typos and grammatical errors as can be identified have been corrected.
We have discussed the results in the "Discussion" section of the paper.
We have clarified the meaning of the statistical outcome measures
We have included a summary of our findings in the conclusion along with thoughts about what should be done about this.
We have corrected Table III